

Article



Effects of Surface Wave-Induced Mixing and Wave-Affected Exchange Coefficients on Tropical Cyclones

Wenqing Zhang ^{1,2,3}, Jialin Zhang ⁴, Qingxiang Liu ^{1,2,3}, Jian Sun ^{1,2,3,*}, Rui Li ³ and Changlong Guan ^{1,2,3}

- ¹ Frontier Science Center for Deep Ocean Multispheres and Earth System (FDOMES), Ocean University of China, 238 Songling Road, Qingdao 266100, China
- ² Physical Oceanography Laboratory, Ocean University of China, 238 Songling Road, Qingdao 266100, China
- ³ College of Oceanic and Atmospheric Sciences, Ocean University of China, 238 Songling Road, Qingdao 266100, China
- ⁴ Qingdao Municipal Marine Development and Promotion Center, 17 Shandong Road, Qingdao 266071, China
- * Correspondence: sunjian77@ouc.edu.cn

Abstract: Surface waves perform a crucial role in modulating tropical cyclone (TC) systems and have proved to be key for numerical TC predictions. In this study, we investigate the effects of wave-induced mixing and wave-affected surface exchange coefficients using a coupled ocean–atmosphere-wave model for two real TC cases: Shanshan (2018) and Megi (2010). The results demonstrate that wave-affected surface exchange coefficients enhance air–sea heat fluxes and have a significantly positive effect on simulated TC intensity, size, and strengthening process. In contrast, the wave-induced mixing has a negative impact on TC intensity and size and is not conducive to TC intensification and maintenance. The net effect of these two factors is a balance between the wave-affected positive contribution and the negative contribution from wave-induced sea surface temperature cooling. We find that the effect of wave-induced mixing on a TC system depends on the local thermal structure of the ocean. When the thermocline is weak and there is warm water at the wave-induced mixing penetration depth, the negative effect of the wave-induced mixing is weak. However, when there is cold subsurface water and a strong thermocline, wave-induced mixing has a significant impact, which exceeds the wave-driven positive feedback.

Keywords: tropical cyclone; wave-induced mixing; wave-affected surface exchange coefficients

1. Introduction

A tropical cyclone (TC) is a powerful atmospheric vortex that can damage human life and industry, so the importance of TC prediction has long been recognized. Over recent decades, significant progress has been made in forecasting TC tracks [1,2]. Between 1990 and 2020, track-forecast errors have reduced by around 60% for forecast periods of 24–120 h, according to National Hurricane Center official forecast error database (https: //www.nhc.noaa.gov/verification/verify7.shtml, accessed on 1 July 2022). However, improvements of TC intensity forecast remain limited [1–3]. The accuracy of TC intensity forecast periods of 24–120 h between 1990 and 2020 (https: //www.nhc.noaa.gov/verification/verify7.shtml, accessed on 1 July 2022). The forecast accuracy for TC size (usually defined as wind radii for a specific wind speed, e.g., 34 kt, 50 kt, or 64 kt) remains uncertain, despite many studies investigating uncertainty and proposing improved methods of estimating TC size [4–7].

A TC is a complex phenomenon of air–sea–wave interactions, and gains energy from the warm upper ocean to develop and mature. The strong winds associated with a TC generate ocean surface waves and drive ocean circulation, which is accompanied by Ekman pumping. These strong winds, surface waves, and local ocean circulation patterns enhance mixing in the upper ocean and entrain cold water below the thermocline. This reduces sea surface temperature (SST), which markedly impacts ocean–air heat transfer, thus weakening



Citation: Zhang, W.; Zhang, J.; Liu, Q.; Sun, J.; Li, R.; Guan, C. Effects of Surface Wave-Induced Mixing and Wave-Affected Exchange Coefficients on Tropical Cyclones. *Remote Sens.* 2023, *15*, 1594. https:// doi.org/10.3390/rs15061594

Academic Editor: Yukiharu Hisaki

Received: 14 January 2023 Revised: 1 March 2023 Accepted: 13 March 2023 Published: 15 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the TC [8–11]. Wave-induced mixing performs an important role in modulating the upper ocean mixing layer under a TC [10,12–15]. Complicated small-scale processes, such as wave breaking, sea spray, and foam bubbles, occur at the sea surface under high wind conditions. These small-scale processes affect air–sea interactions under TC conditions, significantly affecting surface exchange coefficients and promoting TC intensification [16–19].

The physical processes associated with waves at the air–sea interface are not well represented in current numerical models and improvements are needed. For example, ocean models generally do not explicitly include the wave-induced mixing that is attributable to wave breaking, wave orbital motion, or Langmuir circulation [20–22], and this results in overestimates for SST in summer [22–24]. The surface exchange coefficients in atmosphere models (e.g., the drag and enthalpy exchange coefficients) have high uncertainty at high wind speeds due to our limited understanding of the wave-related microphysics of air–sea interactions, such as wave breaking, sea spray, and foam. This lack of understanding stems largely from the difficulties associated with obtaining observations of an extreme sea state [16,17,25]. The incomplete representation of these processes in numerical models, especially in coupled models, limits the accuracies of TC intensity and size prediction [10,11,26,27].

This paper aims to investigate the effects of wave-induced mixing and wave-affected surface exchange coefficients on TCs by hindcasting two TCs with different natures using a coupled ocean–atmosphere–wave model. Our goals are as follows:

- (1) Investigate how wave-induced mixing and wave-affected surface exchange coefficients affect the track, intensity and size of a TC, and analyze air–sea interactions.
- (2) Study the combined effects of wave-induced mixing and wave-affected surface exchange coefficients on TCs, including how the impacts of these two factors differ.

The remainder of this paper is organized as follows: Section 2 describes theoretical wave-related processes (including wave-induced mixing and small-scale wave-related processes at the air–sea interface) and their effects on TC systems. Section 3 briefly describes the coupled ocean–atmosphere–wave model and details the experiment setup. Section 4 presents the results for two TC cases. Section 5 presents a summary and conclusions.

2. Background

2.1. Wave-Induced Mixing

It is known that surface waves contribute to mixing in the upper ocean [20–22,28–30]. Wave breaking can generate turbulence and directly facilitate mixing down to a depth of the order of the wave height [31,32]. In general, wave breaking has only a small effect on SST unless the thermocline is very shallow [33–35]. Zhang W. et al. [10] investigated the effect of wave-breaking induced mixing on Typhoon Megi (2010) and found that it had quite a small impact on the typhoon intensity and size because the mixing due to wave breaking is limited to the shallow depth near to the surface, which has a minor effect on SST and heat fluxes under relatively deep mixed layer conditions.

Wave orbital motion can also produce turbulence and enhance mixing down to a depth of the order of the wavelength [20,22,36]. Numerical studies have concluded that including wave orbital motion in ocean models can significantly improve predictions of SST and mixed layer depth (MLD) and produce results that are in good agreement with observed physics, such as the thermal structure and profiles of suspended sediment in the upper ocean [10,14,22,36,37]. Recent studies have indicated that wave-orbital-motion induced mixing has a negative effect on TC intensity and size [10,11].

Another mechanism through which surface waves enhance mixing in the upper ocean is Langmuir circulation [21,29,38]. Previous studies have indicated that explicitly implementing Langmuir circulation in ocean models reproduces ocean response to TC conditions well [13,15]. However, there remain large differences among different parameterization schemes for Langmuir circulation, making it difficult to investigate and understand its effects [39].

According to previous studies, it is still an open question that whether the waverelated mixing mechanisms between wave orbital motion and Langmuir circulation are completely different or whether an overlap exists between these two processes, because both of them are directly related to the water particle orbits. The theory of wave orbital motion induced mixing seeks to explain the enhanced turbulence due to the localized shear flow (induced by wave orbits) that transfers energy into turbulence through the way of stretching vortex line [40]. Langmuir circulation or turbulence theory focuses on the effect of Stokes drift interacting with wind induced vorticities, which leads to entrainment and enhanced mixing [21,29]. In this aspect, they seem to be different, but they both involve the interaction between surface waves and other motions. Pragmatically, the difference between these two theories is whether the turbulent mixing changes more directly in response to Stokes drift characteristics or wave orbital velocity statistics. However, it is a challenge to distinguish them clearly, because they are co-vary. Numerical studies show that each of these two wave-related mixing mechanisms can significantly increasing the mixing in the upper ocean when they are considered separately in the ocean model [12–15,36], whereas the parameterization process of these two factors together can be complicated, and there is no reliable parameterization scheme combining these two mechanisms together at present because they are not always distinguishable by comparison with data. In this paper, we temporarily consider wave orbital motion induced mixing to investigate the effect of wave-induced mixing on the TC system, and do not include Langmuir circulation which has various parameterization schemes. We also neglect wave-breaking induced mixing. Zhang W. et al. [10] demonstrated that wave orbital motion performs a greater role in modulating the TC system than does wave breaking, and that the effect of wave breaking is negligible compared with the effect of wave-orbital-motion induced mixing.

Ghantous and Babanin [41] described the turbulence that is driven by wave orbital motion as:

$$P = bk\omega_p^3 \frac{H_S^3}{8} e^{3kz},\tag{1}$$

where *b* is a dimensionless proportionality coefficient, *k* is peak wave number, ω_p is the peak radian frequency, *Hs* is the significant wave height, and *z* is the water depth (positive upward). In this paper, *b* is set to 0.0014, which has been shown to achieve reasonable results that are close to observed values [10,12,42]. *P* can be included in a two-equation turbulence closure scheme, such as the General Length Scale scheme (GLS) [43]:

$$\frac{D}{Dt}(K) = \frac{\partial}{\partial z} \left[\frac{\nu_t}{\sigma_k} \frac{\partial}{\partial z}(K) \right] + P_s + P + P_b - \varepsilon,$$
(2)

$$\frac{D\psi}{Dt} = \frac{\partial}{\partial z} \left(\frac{\nu_t}{\sigma_{\psi}} \frac{\partial \psi}{\partial z} \right) + \frac{\psi}{K} (c_1 (P_s + P) + c_3 P_b - c_2 \varepsilon F_{wall}), \tag{3}$$

where *K* is the turbulent kinetic energy, and ψ is a generic parameter used to describe the turbulence length scale. The first terms on the right hand sides of Equations (2) and (3) are the turbulent transport terms for *K* and ψ , respectively; *P*_s and *P*_b are the shear and buoyancy production, respectively; ε is the dissipation rate of *K*; *F*_{wall} is a wall function; *c*₁, *c*₂, and *c*₃ are constants; *v*_t is the eddy viscosity for momentum; and σ_k and σ_{ψ} are the turbulent Schmidt numbers for *K* and ψ , respectively.

In our study, we have implemented Equation (1) in the code for the ocean model to represent wave-orbital-motion induced mixing (hereafter wave-induced mixing for brevity). The ocean model is described in Section 3.2. Further details of the GLS mixing scheme have been provided by Umlauf and Burchard [43] and Warner et al. [44].

2.2. Wave-Related Small-Scale Processes in the Air–Sea Boundary Layer

Waves at the sea surface undergo complex changes under high wind conditions, including wave breaking and the generation of a large number of sea spray droplets and bubbles. These small-scale wave-related processes can lead to reduction or saturation of the drag coefficient (C_D), which directly affects air–sea momentum fluxes and impacts TC intensity [16,19,45–47]. When sea-spray droplets are injected into the atmosphere, they are accelerated by the wind and gain horizontal momentum. When they fall back to the sea surface, the droplets transfer their horizontal momentum to the ocean. Droplet evaporation processes change the turbulent kinetic energy and buoyancy generation in the air–sea boundary layer, thus changing the stability of the boundary layer. Rastigejev et al. [46] pointed out that the above two processes can reduce the drag coefficient. At the same time, wave breaking leads to changes in the flow characteristics of air blowing over the wave crest, which inhibits steepening of the wave and may reduce or saturate the drag coefficient [16,48,49]. In addition, the presence of foam at high wind speed makes the sea surface smoother and reduces the drag coefficient [19]. In terms of summarizing observations, some measurements have suggested that C_D becomes level-off [16,53–55]. In our present work, we chose a C_D with a saturated trend at high wind speed (details are below), and the examining of a C_D with a decreased trend will be complete in the next work.

Sea spray performs an important role in mediating enthalpy fluxes under TC conditions [9,18,56,57]. When a droplet enters the atmosphere, there is a sensible heat transfer between the droplet and the atmosphere, then evaporation occurs and releases water vapor [49,58]. Evaporation of sea spray droplets can sharply cool the air near the sea surface, thus increasing the interfacial sensible heat flux [59]. Richter and Stern [17] estimated surface enthalpy fluxes obtained from dropsonde data and concluded that ocean–atmosphere enthalpy exchange is likely to be dominated by sea spray when a TC is present.

The enthalpy exchange coefficient (C_K), or of wave-mediated heat fluxes, remain highly uncertain owing to the difficulties of obtaining reliable observations under TC conditions [16,18,60]. By convention, C_K is extrapolated from measurements made at wind speeds of less than 30 m/s for high wind conditions [16]. In the classic and widely used Coupled Ocean–Atmosphere Response Experiment (COARE) algorithm, C_K grows weakly with increasing wind speed [61]. Several studies have indicated that C_K does not depend significantly on wind speed at extremely high wind speeds [17,53]. However, these results are inconclusive, as there are large relative errors or uncertainties in the measurements that have been used, and in SST values that have been obtained using indirect methods [16,17]. Recent laboratory results from Komori et al. [16] and Troitskaya et al. [54] show that C_K increases sharply at wind speeds exceeding 33–35 m/s. Komori et al. [16] suggested that this sharp increase in C_K is caused by wave breaking and sea spray. Troitskaya et al. [60] hypothesized that this phenomenon may be associated with increased whitecapping.

At high wind speeds, the surface exchange coefficients, C_D and C_K , are dominated by wave-related small-scale processes, as discussed earlier. We refer to these coefficients as wave-affected surface exchange coefficients. The values for C_D and C_K that are proposed by Komori et al. [16] were measured under 10 m wind speeds ranging from 7.3 m/s to 67 m/s in a high-speed wind–wave tank and are expressed as:

$$C_D = \begin{cases} 1.0 \times 10^{-3} & U_{10} < 5.2 \text{ m/s} \\ 4.4 \times 10^{-4} U_{10}^{0.5} & 5.2 \le U_{10} < 33.6 \text{ m/s}, \\ 2.55 \times 10^{-3} & U_{10} \ge 33.6 \text{ m/s} \end{cases}$$
(4)

$$C_{K} = \begin{cases} 1.39 \times 10^{-3} & U_{10} < 33.6 \text{ m/s} \\ 6.51 \times 10^{-5} U_{10} - 7.99 \times 10^{-4} & U_{10} \ge 33.6 \text{ m/s} \end{cases}$$
(5)

These two formulations for C_D and C_K implicitly include wave-related small-scale processes, such as wave breaking, sea spray, and foam. In this paper, we use Equations (4) and (5) to define wave-affected exchange coefficients and to calculate air–sea momentum and heat fluxes using the following bulk algorithm:

$$\tau = \rho_a C_D U_{10}^2,\tag{6}$$

$$Q_L = -\rho_a L_v C_L U_{10}(q_{10} - q_s), \tag{7}$$

$$Q_{S} = -\rho_{a}C_{p}C_{S}U_{10}(\theta_{10} - \theta_{s}),$$
(8)

where τ , Q_L , and Q_S are the momentum, latent heat, and sensible heat fluxes, respectively; ρ_a is air density; L_v is the latent heat of vaporization; C_p is the specific heat capacity of humid air; C_D , C_L , and C_S are the surface exchange coefficients at 10 m height for drag (momentum), latent heat, and sensible heat, respectively; U_{10} is the 10 m wind speed above the sea level; q is the specific humidity; and θ is temperature. Subscripts "10" and "s" indicate reference heights of 10 m and the sea surface, respectively. Komori et al. [16] verified the conventional assumption that $C_K = C_L = C_S$. Figure 1 shows the curves for C_D and C_K from COARE 3.0 (for details, see the Appendix A) and Komori et al. [16].



Figure 1. Variations of drag coefficient (C_D) and enthalpy exchange coefficient (C_K) with U_{10} .

In some studies into the effect of sea spray on TCs, spray-mediated heat fluxes are added to interfacial heat fluxes, which are generally estimated using a value for C_K that is extrapolated from measurements made at low and moderate wind speeds, following convention [9,11,62–64]. In this case, the spray-mediated heat fluxes depend on the sea spray generation function (SSGF). However, the SSGF is uncertain because of the difficulties of recording observations under the severe conditions of TCs, meaning that the appropriate form for the SSGF remains controversial [57]. Different SSGFs predict droplet amounts that vary by more than six orders of magnitude, assuming a consistent droplet size and wind speed [57,65]. Recently, Sroka and Emanual [18] suggested that calculation of spraymediated flux can be represented more simply as a function of sea spray volume, making the calculated flux insensitive to the shape of the SSGF. However, different SSGFs result in vastly different sea spray volume flux estimates [18], thus leading to large uncertainties in calculations of the heat flux under sea spray conditions.

3. Materials and Methods

3.1. Experiment Design

In this study, we chose two real TC cases with different intensities to evaluate the impacts of wave-related small-scale processes and wave-induced mixing on TCs. The first

case is Typhoon Shanshan (2018), which had a relatively weak intensity, with peak maximum sustained winds of around 36 m/s, according to the Japan Meteorological Agency (JMA). Shanshan originated as a tropical depression in the western North Pacific at 08:00 UTC on 3 August 2018, then moved north-northwestward and strengthened gradually (Figure 2). Shanshan was upgraded to typhoon grade (maximum sustained winds of \geq 32 m/s) on 4 August and maintained maximum sustained winds near 36 m/s for the next five days. Shanshan's center made a close approach to Japan's eastern coast and travelled roughly parallel to the Japanese coasts of Chiba and Ibaraki prefectures on 8 and 9 August. On the morning of 9 August, Shanshan made a sharp turn to the east-northeast and accelerated into the open waters of the North Pacific, where its intensity began to weaken. Finally, Shanshan ceased to be numbered at 14:00 UTC on 10 August 2018.

Expts.	Parameterizations for the Surface Exchange Coefficients	Wave-Induced Mixing
R0	COARE 3.0	No
R1	Equations (4) and (5)	No
R2	COARE 3.0	Equation (1)
R3	Equations (4) and (5)	Equation (1)

Table 1. Numerical experiment design.



Figure 2. JMA best track data for Shanshan and simulated results from four different numerical experiments at 6-hourly intervals from 00:00 UTC 4 August to 00:00 UTC 8 August 2018. (a) Track; (b) minimum sea surface pressure; (c) maximum wind speed at 10 m. Grey lines in panel (a) indicate the ground tracks of altimeters Jason-2, -3, and Sentinel during the period covered in the Shanshan simulations. See Table 1 for details of the different numerical experiments.

The second case is Super Typhoon Megi (2010), which was the strongest tropical cyclone recorded in 2010, with maximum sustained winds up to 65 m/s at its peak intensity, according to JMA. Megi began in the western North Pacific at 00:00 UTC on 13 October 2010, then moved northwest and strengthened gradually (Figure 3). After 12:00 UTC on 16 October, Megi moved gradually west-southwestward and continued to strengthen, then made landfall in the northeastern Philippines at 04:25 UTC on 18 October. After passing through the Philippines, Megi entered the South China Sea and turned sharply to the north at 00:00 UTC on 20 October, lingering in the South China Sea for



five days. Finally, Megi made its second landfall on the southern coast of Fujian in China at 04:55 UTC on 23 October.

Figure 3. JMA best track data for Megi and simulated results from four different numerical experiments at 6 h intervals from 00:00 UTC 15 October to 18:00 UTC 17 October 2010. (a) Track; (b) minimum sea surface pressure; (c) maximum wind speed at 10 m. Grey lines in panel (a) indicate the ground tracks for altimeters Jason-1, -2, Cryosat, and Envisat during the period covered in the Megi simulations. See Table 1 for details of the different numerical experiments.

We conducted four numerical experiments for each TC case to investigate the effects of surface waves (including wave-induced mixing and wave-affected momentum, and enthalpy exchange coefficients) on the TC system. Experiment R0 was the control run, in which we used the classical COARE 3.0 algorithm to calculate surface exchange coefficients (see details in the Appendix A) and excluded the effect of wave-induced mixing. In experiment R1, we implemented the leveled-off drag coefficient from Equation (4) and the high enthalpy exchange coefficient from Equation (5) at high wind speeds. In experiment R2, we used Equation (1) to consider the effect of wave-induced mixing. Finally, the effects of the wave-affected surface exchange coefficients and wave-induced mixing were taken into account in experiment R3. Table 1 presents a summary of the experiments.

3.2. Model Description and Set up

We use the coupled ocean-atmosphere-wave-sediment transport model COAWST [66] to investigate the effects of wave-affected surface exchange coefficients and wave-induced mixing on the two TC systems. COAWST includes atmosphere, ocean, wave, and sediment components, i.e., the Advanced Research Weather Research and Forecasting model (ARW-WRF, hereafter WRF) [67], the Regional Ocean Modeling System (ROMS; http: //www.myroms.org, accessed on 2 June 2022), the Simulating Waves Nearshore model (SWAN) [68], and the USGS Community Sediment Model [69], respectively. The latest version of COAWST also includes a hydrology component—Hydrologic Rainfall-Runoff Model (WRF-Hydro, https://ral.ucar.edu/projects/wrf_hydro/overview, accessed on 2 June 2022), a state-of-the-art spectral wave modeling system—WAVEWATCH III [70], and an infragravity wave driver—Infragravity wave model (InWave). The three components

of COAWST used in our numerical experiments (WRF, ROMS, and SWAN) are described briefly below. A full description of COAWST has been provided by Warner et al. [66] and in https://www.usgs.gov/centers/whcmsc/science/coawst-coupled-ocean-atmosphere-wave-sediment-transport-modeling-system#overview (accessed on 2 July 2022).

WRF is a flexible, state-of-the-art atmosphere simulation system that serves a broad range of meteorological applications across scales from tens of meters to thousands of kilometers. It provides a variety of schemes for the representation of different physical processes, such as microphysics, surface layers, and longwave and shortwave radiation schemes. In our numerical experiments, we used the Nakanishi and Niino surface layer scheme [71], the Rapid Radiative Transfer Model longwave radiation scheme [72], the Dudhia shortwave radiation scheme [73], the Lin microphysics scheme [74], and the Noah land surface scheme [75].

ROMS is a three-dimensional (3D), free surface, terrain-following numerical model that is appropriate for (and widely used in) a diverse range of applications including the current system, astronomical tide movement, ecosystem, and sea ice. In ROMS, transport equations are utilized to solve momentum, scalar advection, and diffusion processes, and an equation of state—accounting for temperature, salinity, and suspended-sediment contributions—is used to compute the density field. In our study, we utilize the Chapman and Flather conditions for the free surface and two-dimensional momentum, respectively, and we use the radiation boundary condition with nudging for 3D momentum. We use the thirdorder upstream horizontal advection scheme, the fourth-order centered vertical advection scheme, and the harmonic horizontal diffusion scheme for momentum, temperature, and salinity in our experiments, and we adopt the GLS vertical mixing scheme for mixing [43].

SWAN is a third-generation wave model and can be used for coastal regions and inland waters. It accounts for various wave related physics, including wave propagation in time and space, shoaling, refraction owing to current and depth, frequency shifting owing to current and non-stationary depth, wind-driven wave generation, three- and fourwave interactions, whitecapping, bottom friction and depth-induced breaking, dissipation, wave-induced set-up, transmission through and reflection against obstacles, and diffraction.

The Model Coupling Toolkit is used in COAWST to exchange data between the three sub-model components. When WRF is coupled to ROMS, the momentum and heat fluxes calculated by WRF are transferred to ROMS using a flux-conservative remapping scheme which ensures consistent momentum and heat fluxes between WRF and ROMS, and the sea surface temperature is passed from ROMS to WRF for estimation of air-sea heat fluxes. In WRF–SWAN coupling, WRF estimates of 10 m winds are used to drive SWAN, and COAWST provides three options to allow exchange of wave data (significant wave height, wavelength, and wave period) to WRF for calculating the surface roughness through consideration of wave-atmosphere interactions [76]. If none of these three options is selected, the COARE 3.0 will be active. In our study, we conducted the experiment R0 as the control run by activating the COARE 3.0 algorithm, then modified the code (in experiments R1 and R3) to estimate air–sea fluxes using Equations (4) and (5), which implicitly include the effects of wave-related small-scale processes, such as wave breaking, sea spray, and foam. The coupling between ROMS and SWAN allows the impacts of surface waves on ocean circulation to be taken into account in ROMS [77]. Wave-related parameters, such as significant wave height, wavelength, wave direction, wave period (surface and bottom), percentage wave breaking, wave energy dissipation, and bottom orbital velocity, are passed from SWAN to ROMS. Currents and sea surface elevations are passed from ROMS to SWAN for modulation of the wave fields. Note that the C_D scheme in SWAN has not been adjusted to be consistent with WRF in COAWST. We followed the default setting for C_D to avoid some potential risks or conflicts that crash the wave model. In the wave model, there are various empirical parameters, and many of them have been tuned based on the C_D . When the C_D is changed in SWAN, the balance between the original wind input and the wave breaking dissipation could be broken. Therefore, a more rigorous way to do this is to change the C_D , then recalibrate all the empirical parameters of the model again

(e.g., parameters of the wave breaking term), but this is another research topic, which is beyond the scope of our present paper. In terms of the consistency between sub-model components, more work on the coupling model is required.

For Typhoon Shanshan, the simulation period was from 00:00 UTC 4 August to 00:00 UTC 8 August 2018, and the modeled domain was 15°N–36°N, 137°E–154°E. For Super Typhoon Megi, the simulation period was from 00:00 UTC 15 October to 18:00 UTC 17 October 2010, and the modeled domain was 10°N–24°N, 118°E–141°E. ROMS was initialized from a hot start following the solutions provided by previous simulation works that simulated TCs using COAWST [66,78,79]. The outputs of Hybrid Coordinate Ocean Model with the Navy Coupled Ocean Data Assimilation (NCODA) system (hereafter HYCOM GLBa0.08; see details in Section 3.3), such as 3-D velocity, temperature, salinity, and sea surface elevation, were used to derive the initial and boundary conditions for ROMS. This data quality is sufficient for our simulation of several days. Additionally, the Orlanski-type radiation condition was used together with relaxation to downscale daily salinity and temperature tracer, and 3-D velocity of HYCOM GLBa0.08. Boundary conditions of the free-surface and depth-averaged velocity were solved using Flather scheme with HYCOM GLBa0.08 data. The spin up periods for WRF and SWAN were both 24 h, so the results of the first day of the simulation periods were not considered in the validation part. WRF started with data from the National Centers for Environmental Prediction (NCEP) Final Operational Global Analysis (FNL) dataset and the initial field was also adjusted by the JMA best track data through the use of a vortex scheme embedded in WRF. SWAN was initialized using the stationary mode to obtain a steady state. A uniform grid spacing of 6 km is used for both typhoon cases, and the same horizontal grids are used for ROMS, WRF, and SWAN $(273 \times 381 \text{ horizontal grids for Shanshan and } 424 \times 256 \text{ horizontal grids for Megi})$. ROMS contains 30 vertical levels, described by the stretched terrain-following coordinate, with higher vertical resolution in the upper mixing layer; WRF uses 31 sigma levels. SWAN resolves 24 wave frequencies (logarithmically spaced from 0.04 to 1.0 Hz) and 36 wave direction bands, each 10° wide. The time steps for ROMS, WRF, and SWAN are 30 s, 30 s, and 60 s, respectively. The three model components exchange data every 10 min.

3.3. Data Description

The initial and lateral boundary conditions for WRF (e.g., air temperature, pressure, winds, and humidity) were taken from the National Centers for Environmental Prediction (NCEP) Final Operational Global Analysis (FNL) dataset (https://rda.ucar.edu/datasets/ds083.2, accessed on 2 June 2022), with spatial and temporal resolutions of 1° and 6 h, respectively. The initial and boundary conditions of ocean temperature, salinity, current, and sea surface elevation for ROMS were obtained from the HYCOM GLABa0.08 dataset (https://www.hycom.org/dataserver, accessed on 2 June 2022), which has a spatial resolution of 1/12° and is assimilated with available SST from satellite, and available vertical temperature and salinity profiles from XBTs, ARGO floats, and moored buoys. The boundary conditions for SWAN were taken as JONSWAP spectra [80].

The best track data for Shanshan and Megi from the JMA (https://www.jma.go.jp/ jma/jma-eng/jma-center/rsmc-hp-pub-eg/trackarchives.html, accessed on 2 June 2022) were used to verify our simulation results. This dataset covers tropical cyclones that have developed over the western North Pacific since 1951. The present version of the dataset includes 6-hourly tracking, intensity, and size analyses. The daily microwaveonly (MWO) SST products from the Remote Sensing System (RSS; https://www.remss. com/measurements/sea-surface-temperature/, accessed on 2 June 2022) were used to assess the modeled sea surface temperature output from the ocean model. The wind data from the Advanced Scatterometer (ASCAT) onboard the EUMETSAT MetOp-A platform (https://www.remss.com/missions/ascat/, accessed on 2 June 2022) and the WindSat radiometer onboard the Coriolis satellite (https://www.remss.com/missions/windsat/, accessed on 2 June 2022) were used to validate our simulated TC size. Significant wave heights and wind speeds derived from six altimeters—Jason-1, -2, -3, Sentinel, Cryosat, and Envisat (sourced from Australia's Integrated Marine Observing System)—were used to assess our modeled wave heights and wind speeds (http://portal.aodn.org.au/search, accessed on 2 June 2022) [81].

4. Results

4.1. TC Track, Intensity, and Size 4.1.1. TC Track

Figures 2a and 3a present the simulated and best tracks for Shanshan and Megi, respectively. These show that the modeled tracks generally match the best tracks well. For Shanshan, the simulated tracks for experiments R1 and R3, which consider the wave-affected surface exchange coefficients, have large errors, relative to experiments R0 (the control run) and R2 (which only includes wave-induced mixing). For Megi, the track errors in experiments R1 and R3 are smaller than those in experiments R0 and R2. The track errors are listed in detail in Table 2. The differences between R0 and R2 are small. These track results suggest that, for both TCs, the wave-affected surface exchange coefficients have a greater effect on the simulated TC track than does wave-induced mixing.

Table 2. Statistics for the track, maximum wind speed at 10 m, minimum sea surface pressure (SSP), and longest radius of 30 kt winds, comparing modeled and best track data. See Section 3.1 for details of the different experiments.

Shanshan										
Expt.	Track	Min SSP		Max Winds at 10 m		Longest Radius of 30 kt				
	Error (km)	Error (hPa)	Relative Error (%)	Error (m/s)	Relative Error (%)	Error (km)	Relative Error (%)			
R0	52.7	-6.7	0.75	0.6	4.5	-49.1	8.8			
R1	111.0	-16.2	1.67	6.8	19.0	-5.8	3.4			
R2	47.7	0.2	0.45	-3.1	8.7	-103.2	18.6			
R3	91.6	-4.2	0.64	-0.8	5.2	-90.2	16.2			
Megi										
Expt.	Track	Min SSP		Max Winds at 10 m		Longest Radius of 30 kt				
	Error (km)	Error (hPa)	Relative Error (%)	Error (m/s)	Relative Error (%)	Error (km)	Relative Error (%)			
R0	106.2	42.9	4.7	-11.8	23.3	-59.1	13.8			
R1	98.6	37.2	4.0	-8.4	17.0	-19.1	7.9			
R2	110.1	44.3	4.8	-13.9	27.1	-74.7	17.4			
R3	93.0	40.9	4.4	-10.7	21.4	-39.7	9.4			

4.1.2. TC Intensity

Figure 2b,c depicts the time series for intensity (the minimum sea surface pressure [SSP], and maximum wind speed at 10 m) for Shanshan. The best track data show that Shanshan intensified between 00:00 UTC and 18:00 UTC on 4 August 2018, then maintained a maximum sustained wind speed close to 36 m/s. In R0, the intensity is slightly overestimated with small biases (Table 2) in the simulated minimum SSP (and maximum winds at 10 m), relative to the best track data. When the wave-affected surface exchange coefficients are used in R1, the intensity of Shanshan is strongly overestimated, with a lower (or smaller) minimum SSP and higher maximum wind speeds at 10 m than the best track data. The simulated maximum wind speed at 10 m (and minimum SSP) in R1 is excessively strengthened (deepened), reaching a relatively high wind speed of 50 m/s (low pressure of 945 hPa) at 06:00 UTC on 6 August 2018, then weakens (rises) gradually. In R2, which includes wave-induced mixing, the maximum wind speed at 10 m rises more slowly and reaches typhoon grade later than in the best track data, then weakens. Wave-induced mixing generally weakens the TC intensity; this is because of the enhanced

s to cooler SSTs and ultimately reduces

mixing, driven by wave orbital motion, which leads to cooler SSTs and ultimately reduces the heat flux from the ocean to the air (see detailed discussion in Sections 4.3 and 4.4). In R3, both the wave-affected surface exchange coefficients and wave-induced mixing are taken into account. The results for R3 show that the combined wave-related effects result in a relatively small error for the minimum SSP and maximum winds at 10 m compared with the best track data.

The intensities simulated for Megi are weaker than the best track data for all four experiments (Figure 3 and Table 2). Comparison with R0 shows that the wave-affected surface exchange coefficients in R1 have a positive effect on the simulated intensity, leading to the deepest minimum SSP and the strongest maximum winds at 10 m among the four simulations. The wave-induced mixing in R2 has a negative impact on the simulated intensity, relative to R0, and results in the shallowest minimum SSP and weakest maximum winds at 10 m. The intensity in R1 strengthens more quickly and follows the observations more closely than in the other experiments. In R2, wave-induced mixing leads to slow development of the minimum SSP and maximum 10 m winds (same scenario in Shanshan), suggesting that wave-induced mixing is not conducive to TC intensification and maintenance. In R3, the impacts of the two wave-related processes counter each other and the combined effect is neutral.

Taken overall, the results for track and intensity show that the coupled model with only wave-induced mixing seems best for Shanshan, but the model with only waveaffected coefficients performs best for Megi. This may be due to the intrinsic limitation of dynamical model for TC simulations. Current dynamical models for TC intensity predictions or simulations have exhibited a tendency to over intensify weak storms and under intensify strong systems due to our insufficient understanding of the coupled airsea-wave system [82]. For Shanshan, which has a weak intensity, the wave-induced mixing performs a critical role in intensity predictions and the intensity forecasts are improved with the effect of wave-induced mixing included in the model. For Megi, which has an extremely high intensity, the original model (R0) does not perform well. The waveaffected coefficients including the wave-related small-scale processes are conducive to TC intensification by enhancing the air-sea heat fluxes, but the intensity forecasts with the wave-affected coefficients are still weaker than the best track data. It is likely that limitations in the model physics will have to be adequately resolved before skillful intensity forecasts become routine for strong or super typhoons. Another possible source of error in the intensity forecasts or simulations for these two typhoon cases is the model settings including the spatial and temporal resolution, and the frequency of data exchange between the three sub-model components.

4.1.3. TC Size

At 08:12 UTC on 6 August 2018, the WindSat radiometer recorded 10 m wind speed observations for Shanshan (Figure 4a). Figure 4b,c,e,f show the simulated 10 m winds for our four experiments. The black lines in these panels represent the radius of 15.4 m/s (30 kt) winds (defined by JMA as describing the TC size). We note that data from radiometers usually have large uncertainties at high wind speeds (>20 m/s) due to the lack of reliable ground observations for calibration (and validation) and the impact of the rain that is generally associated with a TC [83,84]. Therefore, we use WindSat wind speed observations that are below 20 m/s to validate the simulated pattern for the TC size. As shown in Figure 4b,c,e,f, all the simulations indicate that the radius on the right side of the track direction (the eastern side of the TC center) is larger than that on the left side (the western side of the TC center), which is consistent with the observations. However, the TC size is overestimated in R1 relative to the observations, whereas the modeled size in R2 is underestimated. Compared with the best track data, the simulated TC size for the longest radius of 15.4 m/s winds in R1, which includes the wave-affected coefficients, agrees well with the observation (Figure 4d), with only very small errors (Table 2). In contrast, R2, which considers the wave-induced mixing, leads to quite a large underestimation for the

longest radius of 15.4 m/s winds, relative to the best track data. Similar to R2, in the results from R3 (which include both wave-related effects), the simulated radius is generally underestimated and much smaller than in R0. In our comparison of the simulated and best track TC sizes, we focus on the longest radius of 15.4 m/s wind speed, as this parameter represents the region of maximum influence for a TC and is used for TC monitoring, warnings, cost–benefit analyses, and damage assessments.



Figure 4. Shanshan wind speeds at 10 m above the sea surface. (**a**) WindSat wind speed observations at 08:12 UTC on 6 August, 2018; (**b**,**c**,**e**,**f**) simulated wind speeds from four different numerical experiments at 08:00 UTC on 6 August, 2018; (**d**) longest radius of 30 kt (15.4 m/s) from JMA best track data and four different numerical experiments at 6-hourly intervals from 00:00 UTC 4 August through 00:00 UTC 8 August 2018. Black lines in panels (**a**–**c**,**e**,**f**) represent the 15.4 m/s isotach. See Table 1 for details of the different numerical experiments.

The ASCAT scatterometer observed the wind fields of Megi at 13:12 UTC on 17 October 2010 (Figure 5a), and we use these observations to assess the pattern of TC size (radius of 15.4 m/s winds) for Megi. As for the WindSat radiometer, the ASCAT scatterometer cannot provide reliable wind speed data under high wind speed conditions (greater than 25 m/s) [85,86]. The ASCAT data in Figure 5a show that the TC size on the right side of the track direction (the northern side of the TC center) is larger than that on the left side (the southern side of the TC center), despite only half of the wind field were caught up. The TC sizes simulated in all four experiments capture the patterns seen in the ASCAT observations (Figure 5b,c,e,f). Compared with the best track data, we found that the simulated longest radius for 15.4 m/s was underestimated by different amounts in all four simulations (Table 2). The bias is smallest in R1, where the simulated radius is only 7.9% lower than the best track data. In R0, R2, and R3, the underestimation is much greater (Table 2).





Figure 5. Megi wind speeds at 10 m above the sea surface. (a) ASCAT wind speed observations at 13:12 UTC on 17 October 2010; (**b**,**c**,**e**,**f**) simulated wind speeds from four different numerical experiments at 13:00 UTC on 17 October 2010; (d) longest radius of 30 kt (15.4 m/s) from JMA best track data and four different numerical experiments at 6-hourly intervals from 00:00 UTC 15 October through 18:00 UTC 17 October 2010. Black lines in panels (**a**–**c**,**e**,**f**) represent the 15.4 m/s isotach. See Table 1 for details of the different numerical experiments.

For both Shanshan and Megi, the simulated size results suggest that the wave-affected surface coefficients have a significant positive impact on TC size, whereas the wave orbital induced mixing has a negative effect. The combined effect of wave-affected surface coefficients and the wave orbital induced mixing is different for the two TCs. As Figures 4d and 5d show, the simulated longest radii of 15.4 m/s in R3 are smaller than those in R0 for Shanshan, but they are greater for Megi. Similar results are seen for the simulated intensity (Figures 2c and 3c). The modeled maximum wind speeds at 10 m (and the minimum SSP) in R3 are lower (shallower) than those in R0 for Shanshan, but they are higher (deeper) than in R0 for Megi. These results suggest that wave-induced mixing effects dominate over the effects of the wave-affected exchange coefficients for Shanshan, and that the opposite is true for Megi. Comparing the different wave-induced mixing effects simulated for Shanshan and Megi, we see that, when wave-induced mixing is considered in R2, the intensity (size) of Shanshan weakens by 10.4% (11.1%) compared with R0; however, the intensity (size) of Megi reduces by only 5.0% (4.1%) relative to R0. The simulated TC intensity and size are sensitive to SST [9,10,66], and the SST cooling due to wave-induced mixing depends on the vertical temperature structure of the ocean (discussed in Section 4.3). In the case of Shanshan, wave-induced mixing results in a strong decrease in SST owing to the shallow MLD and cold subsurface water (below mixed layer). This leads to a significant reduction in air-sea heat fluxes and thus weakens the TC intensity (and size). In contrast, the effect of wave-induced mixing is relatively small for Megi due to the locally warm subsurface water. This is discussed in more detail in Sections 4.3 and 4.4.

4.2. Validation for Surface Waves and Winds Using Altimeter Observations

In this subsection, we validate the model output for 10 m wind speeds and surface waves using altimeter observations. For Shanshan, we use Jason2, Jason3, and Sentinel Ku band data, and we use Cryosat, Envisat, Jason1, and Jason2 Ku band data for Megi. The ground tracks for these altimeters are shown by grey lines in Figures 2a and 3a. Figures 6 and 7 exhibit comparisons between the modeled and observed 10 m wind

14 of 26

speeds for Shanshan and Megi, respectively. In general, the agreement is good and the simulations are strongly correlated with the observations for all experiments. However, when only the wave-affected surface coefficients are introduced (R1), the 10 m wind speeds are overestimated with a positive mean error (ME). When only wave-induced mixing is considered (R2), the 10 m wind speeds are slightly underestimated. When both the wave-affected surface coefficients and wave-induced mixing are taken into account, the overestimate in R1 and the underestimate in R2 are reduced.

Figures 8 and 9 display the comparison between the simulated and altimeter-observed significant wave height (SWH) for Shanshan and Megi, respectively. These show that the model results agree well with the observations, with a strong correlation. The bias between the modeled SWH and the measurements is between 0.57 m and 1.09 m, indicating that SWH is overestimated in all experiments. Olabarrieta et al. [76], Wu et al. [87], and Zhang W. et al. [10] obtained similar results in studies on TC using COAWST. Olabarrieta et al. [76] attributed this to the model spatial resolution, the model accuracy in different regions, and the method used to derive SWH from the altimeter data. Zhang W. et al. [10] suggested that the bias could be related to the imperfection of wave dissipation source terms in the wave model. The simulated SWHs in R1 are higher than those in R0, reflecting the enhancement of 10 m wind speeds in R1 due to the effect of the wave-affected surface coefficients. In contrast, the wave-induced mixing in R2 results in lower wind speeds than in R0, leading to smaller SWHs in R2 than in R0. However, the reduction in SWH in R2 relative to in R0 is not significant for Megi.



Figure 6. Typhoon Shanshan. Comparison of altimeter-observed 10 m winds with model results from experiments (**a**) R0, (**b**) R1, (**c**) R2, and (**d**) R3. See Table 1 for details of the different numerical experiments.



Figure 7. Typhoon Megi. Comparison of altimeter-observed 10 m winds with model results from experiments (**a**) R0, (**b**) R1, (**c**) R2, and (**d**) R3. See Table 1 for details of the different numerical experiments.



Figure 8. Typhoon Shanshan. Comparison of altimeter observed SWH with model results from experiments (**a**) R0, (**b**) R1, (**c**) R2, and (**d**) R3. See Table 1 for details of the different numerical experiments.



Figure 9. Typhoon Megi. Comparison of altimeter observed SWH with model results from experiments (**a**) R0, (**b**) R1, (**c**) R2, and (**d**) R3. See Table 1 for details of the different numerical experiments.

4.3. SST

In this subsection, we analyze the simulated SST results and verify them through comparison with RSS observations. Figure 10 displays SST patterns on 6 August 2018 for Shanshan for the different experiments and the RSS observations. It shows that the SSTs simulated in R2 and R3, both of which consider wave-induced mixing and agree closely with the observations. SST is overestimated in R0, with a bias of ~0.6 °C. This bias is slightly compensated for R1, which considers wave-affected exchange coefficients that lead to higher wind speed. The SSTs simulated for Megi in R2 and R3 match the RSS data reasonably well (Figure 11). Similar to the simulated SSTs for Shanshan in R0 and R1, the SST simulated for Megi in R0 and R1 are ~0.5 °C and 0.4 °C higher than the observed SST, respectively.



Figure 10. Daily mean SST on 6 August 2018 for Shanshan. (a) SST from RSS observations; (**b**,**c**,**e**,**f**) simulated SSTs from R0, R1, R2, and R3, respectively; (**d**) time series of SSTs averaged over 300 km radius inner core region along Shanshan center location.



Figure 11. Daily mean SST on 17 October 2010 for Megi. (a) SST from RSS observations; (b,c,e,f) simulated SSTs from R0, R1, R2, and R3, respectively; (d) time series of SSTs averaged over 300 km radius inner core region along Megi center location.

Comparing the four modeled SST results shows that, when wave-related small-scale processes are considered using the wave-affected exchange coefficients in R1, the SST is only ~0.1 °C smaller than in R0, despite the fact that considering the wave-affected exchange coefficients results in higher wind speeds. The decrease in SST is stronger in R2 when wave-induced mixing is included. However, the magnitude of the SST reduction due to wave-induced mixing differs between Shanshan and Megi (Figures 10d and 11d). Figure 10d shows that, the SST averaged over 300 km radius inner core region along Shanshan center location, is more strongly reduced in R2 (~1.0 °C) than in R0. The decrease in SST, averaged over 300 km radius inner core region along Megi center location, is weaker in R2 than it is for the case of Shanshan (~0.2 °C; Figure 11d). The SST reduction (due to wave-induced mixing) over the whole simulation domain also shows the similar behavior in these two TC cases (figure not shown due to space constraints).

The different magnitudes for the SST decrease due to wave-induced mixing in our two TC cases are mainly due to differences in the vertical temperature structure of the ocean in the two cases. As shown in Figure 12, the mixed layer in the Shanshan simulations is shallow before the arrival of Shanshan, with an MLD of ~15 m. Below the mixed layer is a strong thermocline. Megi occurred to the south of Shanshan, and the mixed layer simulated for Megi was deep (~40) prior to the arrival of Megi, and the thermocline under Megi is relatively weak. Despite the differences in their MLDs and thermoclines, the SSTs are similar for the Shanshan and Megi simulations prior to the arrival of the TCs (29.0 °C and 29.6 °C, respectively). The wave states (wave height, wave period, and wavelength) are also similar for Shanshan and Megi (figure for wave state is not shown due to space constraints). The penetration depth for wave-induced mixing is of the order of the wavelength [20]. Under TC conditions, the wavelength may reach approximately 100 m for a wave period of 12 s, so the turbulence generated by surface waves penetrates deeply and entrains colder water from below the mixed layer base or thermocline, leading to SST cooling. At a depth of 100 m (the order of wavelength), the water temperature is 25.5 °C in Megi and cooler (19.5 °C) in Shanshan. This suggests that the water entrained from below the mixed layer base (owing to wave orbital motion) is warm in Megi, with a relatively small difference to the SST, so the reduction in SST is small. In contrast, the degree of SST cooling is greater in Shanshan because the temperature difference between the water below the mixed layer and the SST is greater.



Figure 12. Water temperature profiles for Shanshan and Megi at the initial time (prior to the arrival of the TCs) and simulation time of 50 h (the arrival of the TCs) in R0 and R2. The green (red) solid and dotted lines represent the water temperature profiles in R0 (R2) at the simulation time of 50 h for Shanshan and Megi, respectively. The black solid and dotted lines display the water temperature profiles at the initial time for Shanshan and Megi, respectively. Note that the initial water temperature profiles of R0, R1, R2, and R3 for Shanshan (or Megi) are the same.

4.4. Heat Flux

In this subsection, we present the modeled air–sea heat fluxes in different experiments. Unlike previous subsections, here we do not compare the modeled results with other products as there are no available observed resources to be applicable under these two typhoons. The focus here is comparison of heat fluxes across different schemes, and to analyze the effect of wave-related processes on the air–sea heat exchange. Figures 13 and 14a,b show the latent and sensible heat flux distributions and evolutions for Shanshan. When the wave-affected exchange coefficients are used in R1, the latent and sensible heat fluxes increase by 17.5% and 12.0%, respectively, compared with R0. When wave-induced mixing is considered in R2, the latent and sensible heat fluxes decrease significantly by 36.1%, and 39.6%, respectively (relative to R0), owing to the SST cooling that is driven by wave-induced mixing under strong thermocline conditions (discussed in Section 4.3). In R3, when the wave-affected exchange coefficients and wave-induced mixing are both considered, the latent and sensible heat fluxes reduce by 25.7% and 26.2%, respectively (relative to R0), suggesting that wave-induced mixing performs a more dominant role than wave-affected exchange coefficients in modulating heat transfer for Shanshan.

Figures 14c,d and 15 show the latent and sensible heat flux distributions and evolutions for Megi. Compared with R0, the latent and sensible heat fluxes in R1 are enhanced by 28.4% and 22.1%, respectively. In R2, the latent and sensible heat fluxes reduce slightly (by 3.2% and 3.7%, respectively) owing to the minor SST decrease driven by wave-induced mixing when the subsurface (below the mixed layer) water is warm (see Section 4.3). In R3, when both the wave-affected exchange coefficients and wave-induced mixing are considered, the latent and sensible heat fluxes increase by 24.3% and 14.7%, respectively (relative to R0), implying that the wave-mediated exchange flux (or wave-affected exchange coefficients) performs a greater role in modulating heat flux for Megi than wave-induced mixing.



Figure 13. Distributions of heat fluxes for Shanshan. (**a**–**d**) Distributions of modeled latent heat fluxes for 6 August 2018 in R0, R1, R2, and R3; (**e**–**h**) distributions of modeled sensible heat fluxes for 6 August 2018 in R0, R1, R2, and R3. Grey lines are the simulated tracks for Shanshan. See Section 3.1 for details of the simulations.



Figure 14. Evolution of heat fluxes. (**a**) Latent heat fluxes averaged over the 300 km radius inner core region along Shanshan center location; (**b**) same as (**a**), but for sensible heat fluxes; (**c**) same as (**a**), but for Megi; (**d**) same as (**b**), but for Megi. See Section 3.1 for details of the simulations.



Figure 15. Distributions of heat fluxes for Megi. (**a**–**d**) Distributions of modeled latent heat fluxes for 17 October 2010 in R0, R1, R2, and R3; (**e**–**h**) distributions of modeled sensible heat fluxes for 17 October 2010 in R0, R1, R2, and R3. Grey lines are the simulated tracks for Megi. See Section 3.1 for details of the simulations.

The above results for Shanshan and Megi show that including the wave-affected exchange coefficients results in enhancing the estimated heat fluxes and has a positive impact on TC intensity and size. The wave-affected exchange coefficients implicitly include the wave-related small-scale processes (wave breaking, sea spray, and foam) which increase significantly the air–sea heat exchange under high wind speeds. Note that both COARE 3.0 and formulations of Komori et al. [16] implicitly contain wave-driven processes, but COARE 3.0 does not seem applicable for high wind speeds. For moderate and low wind speeds (greater than about 5 m/s and lower than about 25 m/s), wave steepness or wave age performs a dominant role in the surface roughness [61]; however, the wave-related small-scale processes are not yet apparent. COARE 3.0 is obtained at moderate and low wind speeds, therefore, the wave-related processes (such as wave breaking, sea spray, and foam) dominating the air–sea interface are not considered sufficient. Wave-induced mixing leads to a decrease in the estimated heat fluxes and has a negative effect on TC intensity and size. Note that the degree of wave-induced mixing for modulating the TC system depends on the vertical ocean temperature structure.

5. Conclusions

In this study, the effects of surface waves on TC systems have been investigated through numerical simulations using a coupled ocean–atmosphere–wave model (COAWST) and a comparison with available observations. Two factors were considered: wave-induced mixing and wave-affected exchange coefficients. Wave-induced mixing was parameterized by introducing turbulence production due to wave orbital motion [41] into a two-equation turbulence closure scheme. Wave-affected surface exchange coefficients observed at extremely high wind speeds in a laboratory by Komori et al. [16] were implemented into the coupled model. Typhoon Shanshan (2018) and Super Typhoon Megi (2010) were used as case studies.

It is shown that wave-affected surface exchange coefficients can enhance heat fluxes and have a significantly positive effect on TC intensity and size and are conductive to TC strengthening. In contrast, wave-induced mixing has a negative impact on TC intensity and size and does not promote to TC intensification and maintenance. It is found that the magnitude of the impact of wave-induced mixing on a TC system depends on the local ocean vertical temperature profile. Wave-induced mixing can extend to a depth comparable to the wavelength (~100 m) under TC conditions [20], and the mixing induced by long waves penetrates deeply enough to enhance turbulent entrainment near the mixed layer base, deepening the mixed layer and leading to SST cooling. When the thermocline is strong and the subsurface water below the mixed layer is cold, wave-induced mixing results in strong SST cooling and a significant reduction in heat fluxes, and this effect ultimately dominates over the effects the wave-affected exchange coefficients. When the thermocline is weak and the water below the mixed layer is warm, then wave-induced mixing has a relatively small effect on TC intensity and size, and wave-affected exchange coefficients perform a dominant role in modulating the TC system. Note that as the cases used in this study are limited quantitatively, these conclusions are preliminary and subjected to be examined in more TC cases of various properties, such as tropical depressions, decaying TCs, and TCs that move quickly or slowly.

Maybe the experiment (R3) that considers all the effects of surface waves performs not so perfectly, rather we focus on the physical processes of air–sea interaction related to the surface waves and assessing the relative and combined effects of wave-induced mixing and wave-affected surface exchange coefficients on TCs. Although our present model results from R3 seem to have limited improvement compared to R0, we have a clearer and deeper understanding of the complex sea-air-wave interaction under TC conditions. There is still much work to be completed for the improvement of the fully coupled air–sea-wave model system.

Recent studies have shown that, under extreme conditions, such as a TC, the drag coefficient depends on the wave and sea state [64,88]. It is more suitable to use a more complicated parameterization scheme including the wave parameters for the investigation of the effect of wave-affected surface exchange coefficients on TCs. In this study, we did not adopt a wave- or sea-state dependent drag coefficient, instead using the wave-affected exchange coefficients that were derived from laboratory data and are expressed as functions of 10 m wind speed [16] because the available wave- and/or sea-state dependent parameterizations are uncertain due to the difficulty of making observations under TC conditions and the scarcity of field data. More studies with reliable and complicated scheme including the wave parameters are needed in future work when more reliable and complete observations of ocean, atmosphere, and surface waves are obtained. However, we note that laboratory observations do not necessarily reproduce the real environment, and small-scale wave-related processes in the air–sea boundary have not been fully clear and are not explicitly represented. Our study provides a framework and reference for future TC prediction and research.

Author Contributions: Conceptualization, W.Z. and J.S.; formal analysis, C.G.; funding acquisition, W.Z. and J.S.; investigation, W.Z. and J.Z.; methodology, W.Z. and J.Z.; project administration, J.S.; resources, J.S.; validation, W.Z., Q.L. and R.L.; visualization, W.Z. and J.Z.; writing—original draft, W.Z.; writing—review and editing, Q.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the National Natural Science Foundation of China (Grant No. U20A2099, 41906014, 41976017, and 42106012) and National Key Research and Development Program of China (Grant No.2022YFC3105002).

Data Availability Statement: The simulation data used in this study are available at https://doi. org/10.6084/m9.figshare.20449518.v2 (accessed on 1 October 2022) [89]. The NCEP FNL data with 1° × 1° horizontal resolution were downloaded from the NCAR UCAR Research Data Archive Computational and Information System Lab (https://rda.ucar.edu/datasets/ds083.2/, accessed on 2 June 2022) [90]. The initial and boundary conditions for ROMS were taken from the HYCOM GLABa0.08 dataset (https://www.hycom.org/dataserverl, accessed on 2 June 2022) [91]. The best track data were downloaded from JMA (https://www.jma.go.jp/jma/jma-eng/jma-center/rsmchp-pub-eg/trackarchives.html, accessed on 2 June 2022). MW_IR SST, ASCAT, and WindSat data were produced by Remote Sensing Systems and sponsored by NASA [92,93]. These three datasets are available at https://www.remss.com (accessed on 2 June 2022). The significant wave height and wind speeds derived from Jason-1, -2, -3, Sentinel, Cryosat, and Envisat altimeters Ku band were downloaded from the Australian Ocean Data Network (AODN) Portal (http://portal.aodn.org. au/search, accessed on 2 June 2022) [74,94]. Data were sourced from Australia's Integrated Marine Observing System (IMOS)—IMOS is enabled by the National Collaborative Research Infrastructure strategy (NCRIS).

Acknowledgments: The authors thank the researchers who obtained and published the data and schemes used in this study.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

Appendix A. Surface Exchange Coefficients in COARE 3.0

In this paper, we conducted a control experiment (see Section 3.1) for comparison with experiments that considered surface wave effects. We adopted the classical, widely used COARE 3.0 algorithm [61]. In COARE 3.0, C_D , C_L , and C_S have a dependence on surface stability that is determined by the Monin–Obukhov similarity theory (MOST):

$$C_D = \frac{c_{dn}^{1/2}}{\left[1 - \frac{c_{dn}^{1/2}}{\kappa} \Psi_u(\xi)\right]} \frac{c_{dn}^{1/2}}{\left[1 - \frac{c_{dn}^{1/2}}{\kappa} \Psi_u(\xi)\right]},$$
(A1)

$$C_{L} = \frac{c_{ln}^{1/2}}{\left[1 - \frac{c_{ln}^{1/2}}{a\kappa} \Psi_{h}(\xi)\right]} \frac{c_{dn}^{1/2}}{\left[1 - \frac{c_{dn}^{1/2}}{\kappa} \Psi_{u}(\xi)\right]},$$
(A2)

$$C_{S} = \frac{c_{sn}^{1/2}}{\left[1 - \frac{c_{sn}^{1/2}}{a\kappa} \Psi_{h}(\xi)\right]} \frac{c_{dn}^{1/2}}{\left[1 - \frac{c_{dn}^{1/2}}{\kappa} \Psi_{u}(\xi)\right]},$$
(A3)

where κ is the von Karman constant (0.4); Ψ_u and Ψ_h are empirical functions describing the stability dependence of the mean profile for winds and temperature (assumed the same for temperature and humidity), respectively; ξ is the MOST stability parameter; and c_{dn} , c_{ln} , and c_{sn} are exchange coefficients under neutral stability conditions:

$$c_{dn} = \frac{\kappa^2}{\left(\ln(z/z_0)\right)^2},$$
 (A4)

$$c_{ln} = \frac{\kappa^2}{\ln(z/z_0)\ln(z/z_q)},\tag{A5}$$

$$c_{sn} = \frac{\kappa^2}{\ln(z/z_0)\ln(z/z_\theta)},\tag{A6}$$

where z_0 , z_q , and z_θ are the roughness lengths for wind velocity, specific humidity, and temperature, respectively.

In the COARE 3.0 algorithm, z_0 is determined following S. D. Smith [95], using Charnock's [96] relation plus a smooth flow limit:

$$z_0 = \frac{\alpha u_*^2}{g} + \frac{0.11\nu}{u_*},\tag{A7}$$

where α is the Charnock constant; *g* is gravity; *v* is kinematic viscosity; and u_* is friction velocity. In COARE 3.0, α is set to:

$$\alpha = \begin{cases} 0.011 & U_{10} \le 10 \text{ m/s} \\ 0.011 + \frac{0.007}{8}(U_{10} - 10) & 10 < U_{10} < 18 \text{ m/s}, \\ 0.018 & U_{10} \ge 18 \text{ m/s} \end{cases}$$
(A8)

 z_q is calculated using an empirical fit [61]:

$$z_q = \min\left(1.1 \times 10^{-4}, \ 5.5 \times 10^{-5} R_r^{-0.6}\right),$$
 (A9)

where Rr is the roughness Reynolds number. The relationship in Equation (A9) is assumed to also apply for z_{θ} , as $C_K = C_L = C_S$ is assumed.

References

- 1. Cangialosi, J.P.; Blake, E.; DeMaria, M.; Penny, A.; Latto, A.; Rappaport, E.; Tallapragada, V. Recent Progress in Tropical Cyclone Intensity Forecasting at the National Hurricane Center. *Weather Forecast.* **2020**, *35*, 1913–1922. [CrossRef]
- DeMaria, M.; Sampson, C.R.; Knaff, J.A.; Musgrave, K.D. Is Tropical Cyclone Intensity Guidance Improving? Bull. Am. Meteorol. Soc. 2014, 95, 387–398. [CrossRef]
- 3. Emanuel, K.A.; Zhang, F. On the Predictability and Error Sources of Tropical Cyclone Intensity Forecasts. J. Atmos. Sci. 2016, 73, 3739–3747. [CrossRef]
- Chavas, D.R.; Knaff, J.A. A Simple Model for Predicting the Tropical Cyclone Radius of Maximum Wind from Outer Size. Weather Forecast. 2022, 37, 563–579. [CrossRef]
- Knaff, J.A.; Sampson, C.R. After a Decade Are Atlantic Tropical Cyclone Gale Force Wind Radii Forecasts Now Skillful? Weather Forecast. 2015, 30, 702–709. [CrossRef]
- Pun, I.F.; Knaff, J.A.; Sampson, C.R. Uncertainty of Tropical Cyclone Wind Radii on Sea Surface Temperature Cooling. J. Geophys. Res. Atmos. 2021, 126, e2021JD034857. [CrossRef]
- Sampson, C.R.; Goerss, J.S.; Knaff, J.A.; Strahl, B.R.; Fukada, E.M.; Serra, E.A. Tropical Cyclone Gale Wind Radii Estimates, Forecasts, and Error Forecasts for the Western North Pacific. *Weather Forecast.* 2018, 33, 1081–1092. [CrossRef]
- 8. Emanuel, K.A. Thermodynamic control of hurricane intensity. *Nature* 1999, 401, 665–669. [CrossRef]
- 9. Liu, B.; Liu, H.; Xie, L.; Guan, C.; Zhao, D. A Coupled Atmosphere-Wave-Ocean Modeling System: Simulation of the Intensity of an Idealized Tropical Cyclone. *Mon. Weather Rev.* 2011, 139, 132–152. [CrossRef]
- Zhang, W.; Zhao, D.; Zhu, D.; Li, J.; Guan, C.; Sun, J. A Numerical Investigation of the Effect of Wave-Induced Mixing on Tropical Cyclones Using a Coupled Ocean-Atmosphere-Wave Model. J. Geophys. Res. Atmos. 2022, 127, e2021JD036290. [CrossRef]
- 11. Zhao, B.; Qiao, F.; Cavaleri, L.; Wang, G.; Bertotti, L.; Liu, L. Sensitivity of typhoon modeling to surface waves and rainfall. *J. Geophys. Res. Oceans* 2017, 122, 1702–1723. [CrossRef]
- 12. Aijaz, S.; Ghantous, M.; Babanin, A.V.; Ginis, I.; Thomas, B.; Wake, G. Nonbreaking wave-induced mixing in upper ocean during tropical cyclones using coupled hurricane-ocean-wave modeling. *J. Geophys. Res. Oceans* **2017**, *122*, 3939–3963. [CrossRef]
- 13. Reichl, B.G.; Wang, D.; Hara, T.; Ginis, I.; Kukulka, T. Langmuir Turbulence Parameterization in Tropical Cyclone Conditions. *J. Phys. Oceanogr.* **2016**, *46*, 863–886. [CrossRef]
- Toffoli, A.; McConochie, J.; Ghantous, M.; Loffredo, L.; Babanin, A.V. The effect of wave-induced turbulence on the ocean mixed layer during tropical cyclones: Field observations on the Australian North-West Shelf. J. Geophys. Res. Oceans 2012, 117, C00J24. [CrossRef]
- Zhang, X.; Chu, P.C.; Li, W.; Liu, C.; Zhang, L.; Shao, C.; Zhang, X.; Chao, G.; Zhao, Y. Impact of Langmuir Turbulence on the Thermal Response of the Ocean Surface Mixed Layer to Supertyphoon Haitang (2005). *J. Phys. Oceanogr.* 2018, 48, 1651–1674. [CrossRef]
- Komori, S.; Iwano, K.; Takagaki, N.; Onishi, R.; Kurose, R.; Takahashi, K.; Suzuki, N. Laboratory Measurements of Heat Transfer and Drag Coefficients at Extremely High Wind Speeds. J. Phys. Oceanogr. 2018, 48, 959–974. [CrossRef]
- 17. Richter, D.H.; Stern, D.P. Evidence of spray-mediated air-sea enthalpy flux within tropical cyclones. *Geophys. Res. Lett.* 2014, 41, 2997–3003. [CrossRef]
- Sroka, S.; Emanuel, K. Sensitivity of Sea-Surface Enthalpy and Momentum Fluxes to Sea Spray Microphysics. J. Geophys. Res. Oceans 2022, 127, e2021JC017774. [CrossRef]
- 19. Troitskaya, Y.; Sergeev, D.; Kandaurov, A.; Vdovin, M.; Zilitinkevich, S. The Effect of Foam on Waves and the Aerodynamic Roughness of the Water Surface at High Winds. *J. Phys. Oceanogr.* **2019**, *49*, 959–981. [CrossRef]
- 20. Babanin, A.V. On a wave-induced turbulence and a wave-mixed upper ocean layer. *Geophys. Rea. Lett.* 2006, 33, L20605. [CrossRef]
- 21. McWillams, J.C.; Sullivan, P.P.; Moeng, C.-H. Langmuir turbulence in the ocean. J. Fluid Mech. 1997, 334, 1–30. [CrossRef]
- 22. Qiao, F.; Yuan, Y.; Yang, Y.; Zheng, Q.; Xia, C.; Ma, J. Wave-induced mixing in the upper ocean: Distribution and application to a global ocean circulation model. *Geophys. Res. Lett.* **2004**, *31*, L11303. [CrossRef]
- Kantha, L.H.; Clayson, C.A. An improved mixed layer model for geophysical applications. J. Geophys. Res. 1994, 99, 25235–25266. [CrossRef]
- 24. Martin, P.J. Simulation of the mixed layer at OWS November and Papa with several models. J. Geophys. Res. **1985**, 90, 903–916. [CrossRef]
- 25. Green, B.W.; Zhang, F.Q. Impacts of Air-Sea Flux Parameterizations on the Intensity and Structure of Tropical Cyclones. *Mon. Weather Rev.* **2013**, 141, 2308–2324. [CrossRef]

- Chen, S.S.; Price, J.F.; Zhao, W.; Donelan, M.A.; Walsh, E.J. The CBLAST-Hurricane Program and the Next-Generation Fully Coupled Atmosphere-Wave-Ocean Models for Hurricane Research and Prediction. *Bull. Am. Meteorol. Soc.* 2007, *88*, 311–318. [CrossRef]
- 27. Liu, B.; Guan, C.; Xie, L.; Zhao, D. An investigation of the effects of wave state and sea spray on an idealized typhoon using an air-sea coupled modeling system. *Adv. Atmos. Sci.* **2012**, *29*, 391–406. [CrossRef]
- Cavaleri, L.; Fox-Kemper, B.; Hemer, M. Wind Waves in the Coupled Climate System. Bull. Am. Meteorol. Soc. 2012, 93, 1651–1661.
 [CrossRef]
- 29. Li, M.; Zahariev, K.; Garrett, C. Role of Langmuir circulation in the deepening of the ocean surface mixed layer. *Science* **1995**, 270, 1955–1957. [CrossRef]
- 30. Terray, E.A.; Donelan, M.A.; Agrawal, Y.C.; Drennan, W.M.; Kahma, K.K.; Williams, A.J.; Hwang, P.; Kitaigorodskii, S. Estimates of Kinetic Energy Dissipation under Breaking Waves. J. Phys. Oceanogr. **1996**, *26*, 792–804. [CrossRef]
- 31. Babanin, A.V. Breaking and Dissipation of Ocean Surface Waves; Cambridge University Press: Cambridge, UK, 2011.
- Toba, Y.; Kawamura, H. Wind-wave coupled downward-bursting boundary layer (DBBL) beneath the sea surface. *J. Oceanogr.* 1996, 52, 409–419. [CrossRef]
- Burchard, H. Simulating the Wave-Enhanced Layer under Breaking Surface Waves with Two-Equation Turbulence Models. J. Phys. Oceanogr. 2001, 31, 3133–3145. [CrossRef]
- Craig, P.D.; Banner, M.L. Modeling Wave-Enhanced Turbulence in the Ocean Surface Layer. J. Phys. Oceanogr. 1994, 24, 2546–2559. [CrossRef]
- 35. Huang, C.J.; Qiao, F.; Song, Z.; Ezer, T. Improving simulations of the upper ocean by inclusion of surface waves in the Mellor-Yamada turbulence scheme. *J. Geophys. Res.* **2011**, *116*, C01007. [CrossRef]
- Pleskachevsky, A.; Dobrynin, M.; Babanin, A.V.; Günther, H.; Stanev, E. Turbulent Mixing due to Surface Waves Indicated by Remote Sensing of Suspended Particulate Matter and Its Implementation into Coupled Modeling of Waves, Turbulence, and Circulation. J. Phys. Oceanogr. 2011, 41, 708–724. [CrossRef]
- Babanin, A.V.; Ganopolski, A.; Phillips, W.R.C. Wave-induced upper-ocean mixing in a climate model of intermediate complexity. Ocean Model. 2009, 29, 189–197. [CrossRef]
- 38. Smith, J.A. Observed growth of Langmuir circulation. J. Geophys. Res. 1992, 97, 5651–5664. [CrossRef]
- Li, Q.; Reichl, B.G.; Fox-Kemper, B.; Adcroft, A.J.; Belcher, S.E.; Danabasoglu, G.; Grant, A.L.M.; Griffies, S.M.; Hallberg, R.; Hara, T.; et al. Comparing Ocean Surface Boundary Vertical Mixing Schemes Including Langmuir Turbulence. *J. Adv. Model. Earth Syst.* 2019, 11, 3545–3592. [CrossRef]
- 40. Ghantous, M.; Babanin, A.V. Ocean mixing by wave orbital motion. Acta Phys. Slovaca 2014, 64, 1–56.
- 41. Ghantous, M.; Babanin, A.V. One-dimensional modelling of upper ocean mixing by turbulence due to wave orbital motion. *Nonlinear Process. Geophys.* **2014**, *21*, 325–338. [CrossRef]
- 42. Young, I.R.; Babanin, A.V.; Zieger, S. The Decay Rate of Ocean Swell Observed by Altimeter. J. Phys. Oceanogr. 2013, 43, 2322–2333. [CrossRef]
- 43. Umlauf, L.; Burchard, H. A generic length-scale equation for geophysical. J. Mar. Res. 2003, 61, 235–265. [CrossRef]
- 44. Warner, J.C.; Sherwood, C.R.; Arango, H.G.; Signell, R.P. Performance of four turbulence closure models implemented using a generic length scale method. *Ocean Model*. **2005**, *8*, 81–113. [CrossRef]
- Druzhinin, O.A.; Troitskaya, Y.I.; Zilitinkevich, S.S. The Study of Momentum, Mass, and Heat Transfer in a Droplet-Laden Turbulent Airflow Over a Waved Water Surface by Direct Numerical Simulation. J. Geophys. Res. Oceans 2018, 123, 8346–8365.
 [CrossRef]
- Rastigejev, Y.; Suslov, S.A.; Lin, Y.-L. Effect of Ocean Spray on Vertical Momentum Transport Under High-Wind Conditions. Bound. Layer Meteorol. 2011, 141, 1–20. [CrossRef]
- 47. Troitskaya, Y.; Druzhinin, O.; Kozlov, D.; Zilitinkevich, S. The "Bag Breakup" Spume Droplet Generation Mechanism at High Winds. Part II: Contribution to Momentum and Enthalpy Transfer. J. Phys. Oceanogr. 2018, 48, 2189–2207. [CrossRef]
- Kudryavtsev, V.N.; Makin, V.K. Aerodynamic roughness of the sea surface at high winds. *Bound. Layer Meteorol.* 2007, 125, 289–303. [CrossRef]
- Troitskaya, Y.; Kandaurov, A.; Ermakova, O.; Kozlov, D.; Sergeev, D.; Zilitinkevich, S. Bag-breakup fragmentation as the dominant mechanism of sea-spray production in high winds. *Sci. Rep.* 2017, 7, 1614. [CrossRef]
- Powell, M.D.; Vickery, P.J.; Reinhold, T.A. Reduced drag coefficient for high wind speeds in tropical cyclones. *Nature* 2003, 422, 279–283. [CrossRef]
- 51. Holthuijsen, L.H.; Powell, M.D.; Pietrzak, J.D. Wind and waves in extreme hurricanes. *J. Geophys. Res.* **2012**, *117*, C09003. [CrossRef]
- 52. Donelan, M.A. On the decrease of the oceanic drag coefficient in high winds. *J. Geophys. Res. Oceans* **2018**, *123*, 1485–1501. [CrossRef]
- 53. Bell, M.M.; Montgomery, M.T.; Emanuel, K.A. Air–Sea Enthalpy and Momentum Exchange at Major Hurricane Wind Speeds Observed during CBLAST. J. Atmos. Sci. 2012, 69, 3197–3222. [CrossRef]
- Takagaki, N.; Komori, S.; Suzuki, N.; Iwano, K.; Kuramoto, T.; Shimada, S.; Kurose, R.; Takahashi, K. Strong correlation between the drag coefficient and the shape of the wind sea spectrum over a broad range of wind speeds. *Geophys. Res. Lett.* 2012, 39, L23604. [CrossRef]

- 55. Curcic, M.; Haus, B.K. Revised estimates of ocean surface drag in strong winds. *Geophys. Res. Lett.* **2020**, 47, e2020GL087647. [CrossRef]
- 56. Andreas, E.L.; Emanuel, K. Effects of Sea Spray on Tropical Cyclone Intensity. J. Atmos. Sci. 2001, 58, 3741–3751. [CrossRef]
- 57. Peng, T.; Richter, D. Sea Spray and Its Feedback Effects: Assessing Bulk Algorithms of Air-Sea Heat Fluxes via Direct Numerical Simulations. *J. Phys. Oceanogr.* 2019, 49, 1403–1421. [CrossRef]
- 58. Veron, F. Ocean Spray. Annu. Rev. Fluid Mech. 2015, 47, 507–538. [CrossRef]
- 59. Korolev, V.S.; Petrichenko, S.A.; Pudov, V.D. Heat and moisture exchange between the ocean and atmosphere in tropical storms Tess and Skip. *Sov. Meteor. Hydrol.* **1990**, *3*, 92–94.
- 60. Troitskaya, Y.; Sergeev, D.; Vdovin, M.; Kandaurov, A.; Ermakova, O.; Takagaki, N. A Laboratory Study of the Effect of Surface Waves on Heat and Momentum Transfer at High Wind Speeds. *J. Geophys. Res. Oceans* **2020**, *125*, e2020JC016276. [CrossRef]
- 61. Fairall, C.W.; Bradley, E.F.; Hare, J.E.; Grachev, A.A.; Edson, J.B. Bulk Parameterization of Air-Sea Fluxes_ Updates and Verification for the COARE Algorithm. *J. Clim.* **2003**, *16*, 571–591. [CrossRef]
- 62. Andreas, E.L.; Mahrt, L.; Vickers, D. An improved bulk air-sea surface flux algorithm, including spray-mediated transfer. *Q. J. R. Meteorol. Soc.* **2014**, *141*, 642–654. [CrossRef]
- 63. Fairall, C.W.; Kepert, J.D.; Holland, G.J. The effect of sea spray on surface energy transports over the ocean. *Glob. Atmos. Ocean Syst.* **1994**, *2*, 121–142.
- 64. Zhang, L.; Zhang, X.; Perrie, W.; Guan, C.; Dan, B.; Sun, C.; Wu, X.; Liu, K.; Li, D. Impact of Sea Spray and Sea Surface Roughness on the Upper Ocean Response to Super Typhoon Haitang (2005). *J. Phys. Oceanogr.* **2021**, *51*, 1929–1945. [CrossRef]
- 65. Andreas, E.L. A review of the sea spray generation function for the open ocean. In *Atmosphere-Ocean Interactions*; WIT: South-hampton, UK, 2002; Volume 1, pp. 1–46.
- 66. Warner, J.C.; Armstrong, B.; He, R.; Zambon, J.B. Development of a Coupled Ocean-Atmosphere-Wave-Sediment Transport (COAWST) Modeling System. *Ocean Model.* **2010**, *35*, 230–244. [CrossRef]
- Skamarock, W.C.; Klemp, J.B.; Dudhia, J.; Gill, D.O.; Barker, D.; Duda, M.G.; Huang, X.-Y.; Wang, W.; Powers, J.G. A Description of the Advanced Research WRF Version 3. NCAR Tech. Note NCAR/TN-475+STR. 2008; 113p. Available online: https: //opensky.ucar.edu/islandora/object/technotes%3A500/datastream/PDF/view (accessed on 2 June 2022).
- 68. Booij, N.R.R.C.; Ris, R.C.; Holthuijsen, L.H. A third-generation wave model for coastal regions: 1. Model description and validation. *J. Geophys. Res. Oceans* **1999**, *104*, 7649–7666. [CrossRef]
- 69. Warner, J.C.; Sherwood, C.R.; Signen, R.P.; Harris, C.K.; Arango, H.G. Development of a three-dimensional, regional, coupled wave, current, and sediment-transport model. *Comput. Geosci.* **2008**, *34*, 1284–1306. [CrossRef]
- The WAVEWATCH III Development Group (WW3DG). User Manual and System Documentation of WAVEWATCH III TM version 6.07, NOAA/NWS/NCEP/MMAB Technical Note 333. 2019. Available online: https://www.weather.gov/sti/coastalact_ww3 (accessed on 2 June 2022).
- Nakanishi, M.; Niino, H. An Improved Mellor-Yamada Level-3 Model: Its Numerical Stability and Application to a Regional Prediction of Advection Fog. *Bound. Layer Meteorol.* 2006, 119, 397–407. [CrossRef]
- 72. Mlawer, E.J.; Taubman, S.J.; Brown, P.D.; Iacono, M.J.; Clough, S.A. Radiative transfer for inhomogeneous atmospheres: RRTM, availdated correlated-k model for the longwave. *J. Geophys. Res.* **1997**, *102*, 16–663. [CrossRef]
- 73. Dudhia, J. Numerical study of convection observed during the winter monsoon experiment using a mesoscale two-dimensional model. *J. Atmos. Sci.* **1989**, *46*, 3077–3107. [CrossRef]
- 74. Lin, Y.-L.; Farley, R.D.; Orville, H.D. Bulk parameterization of the snow field in a cloud model. J. Clim. Appl. Meteorol. 1983, 22, 1065–1092. [CrossRef]
- 75. Chen, F.; Dudhia, J. Coupling an advanced land surface-hydrology model with the Penn State-NCAR MM5 modeling system. Part I: Model implementation and sensitivity. *Mon. Weather Rev.* **2001**, *129*, 569–585. [CrossRef]
- Olabarrieta, M.; Warner, J.C.; Armstrong, B.; Zambon, J.B.; He, R. Ocean-atmosphere dynamics during Hurricane Ida and Nor'Ida: An application of the coupled ocean-atmosphere-wave-sediment transport (COAWST) modeling system. *Ocean Model.* 2012, 43–44, 112–137. [CrossRef]
- Kumar, N.; Voulgaris, G.; Warner, J.C.; Olabarrieta, M. Implementation of the vortex force formalism in the coupled oceanatmosphere wave-sediment transport (COAWST) modeling system for inner shelf and surf zone applications. *Ocean Model.* 2012, 47, 65–95. [CrossRef]
- Zambon, J.B.; He, R.; Warner, J.C. Tropical to extratropical: Marine environmental changes associated with Superstorm Sandy prior to its landfall. *Geophys. Res. Lett.* 2014, 41, 8935–8943. [CrossRef]
- 79. Zambon, J.B.; He, R.; Warner, J.C.; Hegermiller, C.A. Impact of SST and Surface Waves on Hurricane Florence (2018): A Coupled Modeling Investigation. *Weather Forecast.* 2021, *36*, 1713–1734. [CrossRef]
- Hasselmann, K.; Barnett, T.; Bouws, E.; Carlson, H.; Cartwright, D.; Enke, K.; Ewing, J.A.; Gienapp, H.; Hasselmann, D.E.; Kruseman, P.; et al. Measurements of wind-wave growth and swell decay during the Joint north sea wave project (JONSWAP). *Ergänzung Zur Dtsch. Hydrogr. Zeitschrift.* 1973, 12, 1–95.
- 81. Ribal, A.; Young, I.R. 33 years of globally calibrated wave height and wind speed data based on altimeter observations. *Sci. Data* **2019**, *6*, 1–15. [CrossRef]
- 82. Bender, M.A.; Ginis, I. Real-Case Simulations of Hurricane Ocean Interaction Using A High-Resolution Coupled Model: Effects on Hurricane Intensity. *Mon. Weather Rev.* 2000, 128, 917–946. [CrossRef]

- 83. Meissner, T.; Wentz, F.J. Wind vector retrievals under rain with passive satellite microwave radiometers. *IEEE Trans. Geosci. Remote* **2009**, *47*, 3065–3083. [CrossRef]
- 84. Young, I.R.; Sanina, E.; Babanin, A.V. Calibration and Cross Validation of a Global Wind and Wave Database of Altimeter, Radiometer, and Scatterometer Measurements. J. Atmos. Ocean. Technol. 2017, 34, 1285–1306. [CrossRef]
- Tamizi, A.; Young, I.R.; Ribal, A.; Alves, J.-H. Global scatterometer observations of the structure of tropical cyclone wind fields. *Mon. Weather Rev.* 2020, 148, 4673–4692. [CrossRef]
- Ribal, A.; Tamizi, A.; Young, I.R. Calibration of Scatterometer Wind Speed under Hurricane Conditions. J. Phys. Oceanogr. 2021, 38, 1859–1870. [CrossRef]
- Wu, R.; Zhang, H.; Chen, D.; Li, C.; Lin, J. Impact of Typhoon Kalmaegi (2014) on the South China Sea: Simulations using a fully coupled atmosphere-ocean-wave model. *Ocean Model*. 2018, 131, 132–151. [CrossRef]
- Zhou, X.; Hara, T.; Ginis, I.; D'Asaro, E.; Hsu, J.; Reichl, B.G. Drag Coefficient and Its Sea State Dependence under Tropical Cyclones. J. Phys. Oceanogr. 2022, 52, 1447–1470. [CrossRef]
- Zhang, W. Dataset of "Effects of Surface Wave-Induced Mixing and Wave-Affected Exchange Coefficients on Tropical Cyclones". [Dataset] Figshare. 2022. Available online: https://figshare.com/articles/dataset/Dataset_of_Effects_of_surface_waves_on_tropical_cyclones_/20449518/1 (accessed on 1 October 2022).
- National Centers for Environmental Prediction/National Weather Service/NOAA/U.S. Department of Commerce. NCEP FNL
 operational model global tropospheric analyses, continuing from July 1999 [Dataset]. In Research Data Archive at the National
 Center for Atmospheric Research; Computational and Information Systems Laboratory: Boulder, CO, USA, 2000. [CrossRef]
- Cummings, J.A. Operational multivariate ocean data assimilation. *Q. J. R. Meteorol. Soc.* 2005, 131, 3583–3604. [CrossRef]
 Ricciardulli, L.; Wentz, F.J. *Remote Sensing Systems ASCAT C-2015 Daily Ocean Vector Winds on 0.25 Deg Grid, Version 02.1 [Dataset]*;
- Remote Sensing Systems: Santa Rosa, CA, USA, 2016. Available online: https://www.remss.com (accessed on 2 June 2022).
- Wentz, F.J.; Ricciardulli, L.; Gentemann, C.; Meissner, T.; Hilburn, K.A.; Scott, J. Remote Sensing Systems Coriolis WindSat Daily Environmental Suite on 0.25 Deg Grid, Version 7.0.1 [Dataset]; Remote Sensing Systems: Santa Rosa, CA, USA, 2013. Available online: https://www.remss.com/missions/windsat (accessed on 2 June 2022).
- 94. IMOS. IMOS—SRS Surface Waves Sub-Facility—Altimeter Wave/Wind [Dataset]. Australian Ocean Data Network. 2010–2018. Available online: https://portal.aodn.org.au/search (accessed on 2 June 2022).
- 95. Smith, S.D. Coefficients for sea surface wind stress, heat flux, and wind profiles as a function of wind speed and temperature. *J. Geophys. Res.* **1988**, *93*, 15467–15472. [CrossRef]
- 96. Charnock, H. Wind stress on a water surface. Q. J. R. Meteorol. Soc. 1955, 81, 639–640. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.